January 17, 2017

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 31979 revision.doc).

**Title:** Endoscopic submucosal dissection for early gastric cancer with undifferentiated-type histology: A meta-analysis

**Author:** Chang Seok Bang, Gwang Ho Baik, In Soo Shin, Jing Bong Kim, Ki Tae Suk, Jai Hoon Yoon, Yeon Soo Kim, Dong Joon Kim, Woon Geon Shin, Kyung Ho Kim, Hak Yang Kim, Hyun Lim, Ho Seok Kang, Jong Hyeok Kim, Jin Bae Kim, Sung Won Jung, Se Hyub Kae, Hyun Joo Jang, Min Ho Choi

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 31979

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

First of all, we appreciate the Editor’s and reviewers’ thoughtful and helpful comments. Also, we are pleased to have an opportunity to make this paper to be an even better one and to be accepted with revision, because the Editor and reviewers provided additional important points that we have not realized before.

(1) Reviewer 00051081

We appreciate the reviewer’s thoughtful comments. The detailed corrections are listed below.

"Positive alcohol consumption was defined as those who drink twice or more in a week" + alcohol consumption for the last 3 months: these criteria should be revised or detailed. The definition of "drink" should be addressed including local alcoholic beverages) by amount (i.e: g/day or g/week).

-> Alcohol consumption data were recorded as the standard of Soju or beer consumed per week. Alcohol amount in a bottle of Soju is 54g (360ml) and in a bottle of beer is 12g (355ml) in Korea. The minimal amount in the alcoholics group was 162g/week in Soju and 144g in beer. Therefore, the definition of positive alcohol consumption was corrected as more than 20g/day (colored red).

(2) Reviewer 1551804

We appreciate the reviewer’s helpful comments.

“It would have been very interesting to set a control group of not complicated ulcers”.

1
The best way to elucidate the characteristics of peptic ulcer perforation would be comparison with non-perforated peptic ulcer. However, this study was performed as cross-sectional and descriptive format and too small portion of peptic ulcer perforation was identified compared with non-perforated peptic ulcer making difficult to get statistical legitimacy. For the reasoning of the result of this study, acquisition of well-matched control cohort could be possible in the future study, although lacking enough control in this study. Thank you.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the World Journal of Gastroenterology.

Sincerely yours,

Chang Seok Bang, M.D., Ph.D.
Dept. of Internal Medicine
Hallym University College of Medicine
Chuncheon Sacred Heart Hospital
Sakju-ro, 77, Chuncheon, Gangwon-do
24253, South Korea
Tel.: +82-33-240-5000
Fax: +82-33-241-8064
E-mail: cloudslove@naver.com